Author's response to reviews

Title: Does parenting style matter? - Predictors of suicide attempts in a representative sample of adolescents

Authors:

Carolin Donath (carolin.donath@uk-erlangen.de)
Elmar Graessel (elmar.graessel@uk-erlangen.de)
Dirk Baier (baier@kfn.uni-hannover.de)
Stefan Bleich (bleich.stefan@mh-hannover.de)
Thomas Hillemacher (hillemacher.thomas@mh-hannover.de)

Version: 8 Date: 26 March 2014

Dear Editors,

the following changes were made to the manuscript according to the suggestions of reviewer 2:

1. The paragraph in the "Statistical Analysis" section concerning the estimation of predictor number was shortened as suggested by the reviewer.

2. The paragraph concerning the handling of missing values has been strongly summarized as Reviewer 2 suggested.

3. The limitations section was changed to include the fact that depression diagnoses was not available for analysis.

Sincerely,

Carolin Donath